Author's response to reviews

Title: The Healthy Eating and Lifestyle Programme (HELP) study protocol: an efficacy randomised controlled trial of the HELP intervention compared with enhanced standard care of obese adolescent in the community

Authors:

Deborah Christie (Deborah.Christie@uclh.nhs.uk)
Lee Hudson (Lee.Hudson@uclh.nhs.uk)
Anne Mathiot (a.mathiot@ich.ucl.ac.uk)
Tim J Cole (t.cole@ich.ucl.ac.uk)
Saffron Karlsen (s.karlsen@ucl.ac.uk)
Anthony Kessel (anthony.kessel@hpa.org.uk)
Sanjay Kinra (Sanjay.Kinra@lshtm.ac.uk)
Steve Morris (steve.morris@ucl.ac.uk)
Irwin Nazareth (IN@gprf.mrc.ac.uk)
Ulla Sovio (Ulla.Sovio@lshtm.ac.uk)
Ian Wong (ian.wong@pharmacy.ac.uk)
Russell M Viner (r.viner@ich.ucl.ac.uk)

Version: 2 Date: 27 July 2011

Author's response to reviews:

I am pleased to confirm that the following revisions have been made in this version of the manuscript:

Trial registration number has been included in the abstract.

An 'Authors' Contributions' section has been added at the end of the manuscript, before the reference list.

The order of the sections in the manuscript has been revised so that tables follow the reference list.